Author’s response to reviews

Title: Fulminant infection associated with multiple liver abscesses: a case report and literature review.

Authors:
GULTAKIN BAKIROVA (dr.gulya07@gmail.com)
Abdulrahman Alharthy (a_almshal@hotmail.com)
Silvia Corcione (corcione.silvia@gmail.com)
Waleed Tharwat Aletreby (waleedalebreby@gmail.com)
Francesco Giuseppe De Rosa (francescogiuseppe.derosa@unito.it)
Dimitrios Karakitsos (karakitsosdimitrios@gmail.com)
Ahmed Mady (afmady@hotmail.com)

Version: 1 Date: 31 May 2020

Author’s response to reviews:

Dear Sirs,
As per your recommendations we reviewed article and upgrade information provided.
1. we removed name of institution from manuscript body.
2. in line 80 we described methods used for culturing of samples.
3. in line 88 : we are clinicians and not a microbiologists, so apologising for misunderstanding and intrepretation of antimicrobilal susceptability by old guidelines NCCIS system. We contacted with our laboratory and upgraded intrepretation by CLSI guideline.